

# Stability Analysis of Grid-Forming MMC-HVDC Transmission Connected to Legacy Power Systems

Luís Lourenço, Filipe Perez, Alessio Iovine, Gilney Damm, Renato Monaro,

Maurício Salles

### ▶ To cite this version:

Luís Lourenço, Filipe Perez, Alessio Iovine, Gilney Damm, Renato Monaro, et al.. Stability Analysis of Grid-Forming MMC-HVDC Transmission Connected to Legacy Power Systems. Energies, 2021, 14 (23), 25p. 10.3390/en14238017 . hal-03465507

## HAL Id: hal-03465507 https://centralesupelec.hal.science/hal-03465507

Submitted on 15 Dec 2021

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.





### Article Stability Analysis of Grid-Forming MMC-HVDC Transmission Connected to Legacy Power Systems

Luís F. N. Lourenço <sup>1,2,\*</sup>, Filipe Perez <sup>3</sup>, Alessio Iovine <sup>1,4</sup>, Gilney Damm <sup>5</sup> and Renato M. Monaro <sup>2</sup> and Maurício B. C. Salles <sup>2</sup>

- <sup>1</sup> Laboratoire des Signaux et Systèmes L2S, CentraleSupelec, Université Paris-Saclay, 3, rue Joliot Curie, 91190 Gif-sur-Yvette, France; alessio.iovine@centralesupelec.fr
- <sup>2</sup> Laboratory of Advanced Electric Grids LGrid, Polytechnic School, University of São Paulo, 158 Av. Prof. Luciano Gualberto, Travessa 3, Saint Paul 05508-010, Brazil; monaro@usp.br (R.M.M.); mausalles@usp.br (M.B.C.S.)
- <sup>3</sup> Power Systems Group, Lactec Institute, rodovia BR-116, 8813, Jardim das Americas, Curitiba 80215-090, Brazil; filipe.perez@lactec.org.br
- <sup>4</sup> Centre National de la Recherche Scientifique, CNRS, 3, rue Joliot Curie, 91190 Gif-sur-Yvette, France
- <sup>5</sup> COSYS-LISIS, Université Gustave Eiffel, IFSTTAR, 77447 Marne-la-Vallée, France; gilney.damm@univ-eiffel.fr
- \* Correspondence: lfnlourenco@usp.br or luis.normandialourenco@centralesupelec.fr

Abstract: The power system is going through a change in its very foundations. More and more power converters are being integrated into the electric grid to interface renewable energy resources and in high-voltage direct-current (HVDC) transmission systems. This article presents a discussion on the stability of power systems when HVDC transmission systems based on modular multilevel converters (MMC) are connected in grid-forming (GFM) mode to the legacy power system using concepts of energy functions and Lyapunov stability theory and considering aspects of the interoperability between GFM converter technologies. As a base for the stability analysis, we review the main GFM converter technologies (droop and virtual synchronous machine), highlighting their differences. Then, we present a model using the center-of-inertia formulation for a multi-machine/multi-GFM converter power system representing a close future scenario of power systems where GFM converters might adopt different technologies. To illustrate the theoretical Lyapunov-based stability analysis, simulations performed in Matlab/Simulink showed the behavior of a 12-bus test system during a frequency disturbance that originated from the sudden connection of a load. To reflect the interoperability of different GFM technologies and the power system, scenarios with one single GFM technology and a scenario with mixed technologies were investigated. For the test system considered, the frequency response with fewer oscillations and a higher frequency nadir was obtained when all GFM converters were operated as VSMs that have a higher inertial response contribution.

**Keywords:** power system stability; voltage source converters; modular multilevel converter; gridforming converter; future power system; HVDC; energy function stability analysis

#### 1. Introduction

To face the pressing climate crisis and to counteract its effects, all sectors responsible for the emission of greenhouse gases must make significant changes. In the power sector, one of the main courses of change is through the development of renewable energy sources (renewables) interfaced by power converters like those for wind and solar photovoltaic energy. The interconnection of these renewable energy sources can be done by high-voltage direct-current (HVDC) power transmission systems, given the big distances involved and the compatibility with the offshore environment challenges [1]. Furthermore, HVDC transmission can be considered as viable reinforcements for the legacy power system as embedded HVDC lines [2,3].

The accelerated development of converter-based devices such as HVDC transmission and converter-interfaced renewable generation is causing a profound change into the very



Citation: Lourenço, L.F.N.; Perez, F.; Iovine, A.; Damm, G.; Monaro, R.M.; Salle, M.B.C. Stability Analysis of Grid-Forming MMC-HVDC Transmission Connected to Legacy Power Systems. *Energies* **2021**, *14*, 8017. https://doi.org/10.3390/ en14238017

Academic Editor: Abu-Siada Ahmed

Received: 31 October 2021 Accepted: 23 November 2021 Published: 1 December 2021

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). physics of the power system [4,5]. In this scenario, the power generation is shifted from the pollutant synchronous generators based on fossil fuels to converter-based resources. However, in the present structure of the power system, the operators rely on synchronous generators for ancillary services and the provision of inertia and of reliable power reserves. A power system with decreasing inertia levels allows for greater frequency excursions during disturbances that might trigger frequency-based protection equipment [6] and further contribute to deteriorating the operation of the system as described in the blackout report detailed in [7].

An HVDC transmission line might be interfaced to the grid according to the linecommutated converter (LCC) technology or voltage source converter (VSC) technology. The LCC alternative requires a strong grid to function correctly and is unable to control its reactive power output, requiring external sources to compensate its own necessities [8]. Moreover, the direction of the power flow cannot be changed without reversing the polarity of the terminals [9]. However, since the proposition of the VSC topology known as a modular multilevel converter (MMC) [10–12], VSC-HVDC is becoming a viable option for high-voltage/high-power applications [13]. A VSC-HVDC transmission line is capable of controlling independently active and reactive power in both ends, reversing the power flow without changing the polarity of the terminals and operating in multi-terminal configurations, introducing a flexibility that will be required for the operation in the modern power system [14,15].

Presently, most VSC-based systems commissioned around the world are controlled according to a grid-following (GFL) strategy. This operation strategy consists in synchronizing the voltage output of the converter with the frequency, amplitude, and phase angle of the grid voltage [16,17]. This is achieved through a device called a phase-locked loop (PLL) [18], which requires a voltage waveform to synchronize with. However, because of the weakening of the grid due to the shift of power generation to converter-interfaced resources, this voltage is subjected to higher excursions on frequency, which might result in instabilities in the PLL [15]. Moreover, during a disturbance, a GFL converter behaves as a constant current source for a brief period of time, not using the topology inherent control flexibility to support the grid. However, the growing penetration levels of VSCs connected into the power system place the security and the stability of the system at risk [19].

Hence, for the transition to occur towards a sustainable energy mix, new control alternatives for VSC converters must be developed to maintain the system's stable and reliable operation even with reduced penetration of synchronous generators. The grid-forming (GFM) control strategy is one of the main alternatives being considered [6,20,21]. A VSC controlled by a GFM strategy is capable of controlling the amplitude, phase, and frequency at the point of common coupling [21]; thus, the converter is able to react to changes in the grid instead of the constant power control approach in the GFL scheme.

A wide variety of GFM strategies were proposed in the technical literature: the virtual synchronous machine (VSM) [22,23], the synchronverter [24–26], droop control [27,28], and others that were reviewed in [21]. Specifically in the context of HVDC, ref. [26] proposed the use of a synchronverter for a two-level VSC-HVDC transmission system, where the inverter mimics the behavior of a synchronous generator by using the same electromechanical equations used for the modeling of generators. In [29], the interaction between AC and DC sides of an HVDC controlled as a VSM was investigated. Ref. [30] presents a classification of three droop controlled strategies and highlights their applications to a two-level VSC-HVDC transmission system. In [31], a short circuit analysis in an AC offshore grid with multiple VSC droop-controlled GFM converters is present focusing on the transmission of offshore wind power. In [32], a hybrid converter. As the diode rectifier and a two-level VSC was used offshore as a hybrid GFM control strategy. Finally, in [33], the effect of GFM control on the internal energy stored in the submodule capacitors of the MMC was investigated.

Therefore, stability analysis is of interest given the expected growth in GFM applications in power systems (see [34–37]). In [34], a stability analysis was conducted for GFL and GFM converters connected to a synchronous generator, one at a time. The study established a comparison between both energy functions and transient stability for GFL and GFM converters, compared through an experimental setup. Ref. [35] presents a design-oriented study for the transient stability of VSCs controlled according to four GFM strategies (droop, droop with power filtering, power synchronization control, and VSM). The VSC was considered connected to an electric grid, and the transient stability was shown through an experimental setup. The study also highlights conditions where the system was rendered unstable. In [36], a transient stability study was performed considering different types of GFM devices. The analysis was performed considering that a GFM with inertia (namely, a VSM or a synchronous generator) is connected to a droop-controlled GFM inverter, showing through simulations that the system is stable. Furthermore, ref. [37] performed a stability analysis of a VSM connected to a synchronous generator using the Lyapunov direct method. The study presented an estimated region of attraction according to different sets of parameters. Additionally, the study presented an analysis of a three-bus system with two VSMs connected to a synchronous generator.

These stability analyses are very important to understand the limits and characteristics of the system. Since MMC-HVDC equipped with GFM control strategies might not be all from the same manufacturer, that implies that a technological difference must be possible given the choice of which GFM strategy to follow. The works that we highlighted in the previous paragraph considered that a single GFM technology is present at each study, to the exception of [36]. In those, a two-elements system was introduced, which was able to represent the case of one generator and one converter or the case of two converters. Moreover, the reviewed studies presented only one equivalent GFM converter in their analysis, to the exception of [37] that considered the interconnection of two GFM converters following the same VSM technology. Therefore, one possible improvement in the subject is the development of analysis on power systems with multiple machines, multiple GFM converters, and multiple GFM technologies.

Hence, to contribute with a wider stability analysis for power systems, including multiple GFM converters and multiple GFM technologies, the scope of this work was to perform a Lyapunov-based stability analysis for a multi-machine/multi-GFM converter system representing the power system during the energy transition. To meet this objective, we review the two most mature technologies for GFM converters under consideration (droop and VSM), highlighting the differences and similarities between them. Furthermore, we introduce a model for the multi-machine/multi-GFM converter system using the center-of -inertia (COI) formulation to establish a single reference for the frequencies and angle deviations in the system. From the model, it is possible to conduct the proposed Lyapunov-based stability analysis using energy functions, demonstrating that the frequencies of the system will converge to the frequency of the COI. To illustrate the theoretical results, a 12-bus test system was implemented in Matlab/Simulink comprising two synchronous machines and two GFM converters. The simulations were performed considering scenarios where there was a single GFM technology and for a scenario where there were multiple adopted GFM technologies. These results allowed the interoperability factor between different GFM converter technologies and synchronous generators to be analyzed. Moreover, since we considered that the converters are representing HVDC transmission systems, a review on MMC control schemes is presented to discuss the challenges of the implementation of GFM converter strategies using the MMC topology.

This article is structured as follows: Section 2 presents an overview of the MMC-HVDC control scheme; Section 3 details the modeling of the multi-machine and multi-GFM system; Section 4 presents the stability analysis using Lyapunov theory; Section 5 details the test system used and the simulation results; and, finally, Section 6 brings the conclusions.

#### 2. Grid-Forming and Grid-Following Control Strategies for Modular Multilevel Converters

During the early years of power-electronic-interfaced renewables, the stability of the power system was barely affected by the level of nonsynchronous generation. In case of disturbances, transmission system operators (TSOs) would only require the shutdown of the converter-based power plant, which would reconnect only when the disturbance was solved [38]. For this purpose, a simple GFL control strategy to inject active and reactive power to the main grid was required for the grid-connected operation mode for power converters where a PLL structure was used to guarantee that the converter is synchronised with the main power network. In this case, the frequency and voltage output of the converters are governed by the main network, which can be represented most of the time as an infinite bus (from the point of view of the converters).

Because of the growth in the participation of converter-based devices in the electric grid, system operators had to review the policy that required disconnection during the disturbance. In this scenario, if renewables and HVDC lines are simply disconnected, the resulting power imbalance could amplify the magnitude of the disturbance, leading to a blackout event. Hence, grid codes had to be reviewed, including some grid-supporting features for converter-based resources such as a fault ride-through capability and a frequency response capability. Since synchronous generators are responsible for the majority of power generation, the grid still offers an adequate environment for the adoption of the GFL converters with grid-supporting capabilities.

Nowadays, not only are more and more renewables and HVDC projects being commissioned but fossil-fuel-based synchronous generators are being phased out, paving the way to a sustainable power grid. This new power system scenario is already being reported by system operators around the world in regions like Ireland (EirGrid), Great Britain (National Grid), Texas (Electric Reliability Council of Texas, ERCOT), and Australia (Australian Energy Market Operator, AEMO). The shifting generation from synchronous generators to converter-interfaced resources creates a challenging environment for the operation of traditional GFL control structures. Therefore, a new technology concept is emerging in the GFM converters.

A GFM converter is able to regulate its own frequency and voltage. Hence, this control scheme allows for the GFM converter to interact with the grid and to reflect the disturbances in its operation. Moreover, A GFM converter does not rely on the PLL structure for its synchronization with the electric grid (only for measurements in some schemes). These result in a more robust controller for the converter when the power converter is connected to a weak grid with low inertia levels. A drawback for this strategy is that the outer voltage control loop cascaded with the inner control current loop in power converters has complex tuning and can suffer interference from the external parameters of the network, output converter filters, and PLL measurements. Therefore, robust control techniques are required to maintain a suitable operation, regarding the complexity of the power system [39,40].

#### 2.1. Classification of Control Strategies for Power Converters

A classification of VSC converters is proposed in [41], based on their different application, such as GFM, including two different approaches of GFL named "grid-feeding" and "grid-supporting." The GFM converters are modeled as an ideal AC voltage source with low output impedance, where the voltage amplitude and frequency are directly controlled, rendering the converter capable of providing services to contribute to the stable operation of the system. In this case, GFM converters need an accurate synchronization to realize power sharing among other generators connected in parallel.

GFL grid-feeding converters are modeled as a current source in parallel with a high impedance to represent the constant injection of current. They are usually designed to deliver constant active and reactive powers to the grid, which is achieved by the control of the currents. In this case, the synchronization with the grid is needed, where standalone operation is not possible. Last, the GFL grid-supporting converters are modeled as con-

trolled current sources in parallel with a shunt impedance. They are also modeled as a voltage source. In this case, the currents and voltages of the converters are controlled to improve frequency and voltage levels of the connected grid. Extra ancillary services for grid-supporting may be implemented locally, such as inertia emulation, damping power oscillation, and unbalanced compensation. A general scheme of the three applications of power converters is presented in Table 1 according to [41].

Table 1. Different applications of power converters for AC grids.



The majority of the research efforts, so far, were carried out to model and control GFM converters with traditional two-level VSCs. For HVDC applications, however, the most prominent topology is the MMC, capable of handling high-power/high-voltage applications. Therefore, in the sequel of this section, we discuss the challenges of the implementation of GFM strategies on MMCs.

#### 2.2. Modular Multilevel Converter Model

The model of a 2m + 1 level MMC connected to the electric grid is shown in Figure 1. The MMC is composed of six arms formed by a string of *m* series-connected submodules. In this work, the half-bridge topology for the submodules was considered (see [42,43] for more details on submodule topology). Each leg of the MMC corresponds to one phase of the converter, and it is comprised of two arms.

A substantial number of works were carried out in the modeling of the MMC for HVDC applications (see [13,44–52]). Given the chosen model, one can design a control system for the MMC-HVDC. The control problem associated with the MMC is more complex than the one associated with two- or three-level VSCs [48,53]. The complexity arises from the growth in the number of variables to be controlled compared to the two- and three-level VSCs resulting from the converter circulating current and the internal energy stored in the submodules of the capacitors. The internal dynamics of the converter has been shown to be asymptotically stable in a non-energy control approach [54], but in this case the dynamics of the circulating currents are given by the system's physical parameters. In [55], it was shown that the use of an energy-controlled approach with direct control of the circulating currents could improve the response of the internal converter dynamics with the design of a proper control system. In the remaining part of this section, we discuss the implementations of the energy-controlled approach in both GFL and GFM MMCs.

The internal-dynamics controller's structure is dependent on the model used for the circulating currents and for the energy dynamics. The internal dynamics of the MMC can be modeled in different frames. In [56], the circulating currents were modeled in the *abc* frame, in [33,57] in the  $\alpha\beta0$  frame, while in [47] the model was carried out in a decoupled double synchronous frame, and the model proposed by [52] was on the *dq*0 frame. The choice of the frame can influence the complexity of the controller and the accuracy of representation as well. In the simulations carried out further in this work, the *dq*0 bilinear model proposed by [52] was used for the internal converter dynamics.



Figure 1. Model of an MMC-HVDC inverter terminal connected to the electric grid.

#### 2.3. Grid Following (GFL)

The control block diagram of a GFL MMC according to the energy-based control approach is shown in Figure 2. The controller can be divided into two subsystems: one responsible for controlling the power exchange between the converter and the grid and the other one responsible for controlling the internal dynamics of the converter.



Figure 2. Block diagram of an MMC-based GFL converter in grid-feeding mode.

By assuming a balanced system, the AC current controller can be implemented in the dq frame. It is responsible for tracking the current references provided by an upperlevel control by generating the input contribution  $u_{AC,dq}$ . The active and reactive power controllers are responsible for calculating the references for the AC current controller from the reference values provided by the TSO.

The circulating current control is responsible for tracking the circulating current references that are determined by an upper-level controller, generating the input contribution  $u_{dc,dq0}$ . When using the bilinear model for the internal dynamics, the reference of the *d* component of the circulating current can be used to regulate the energy balance  $\Delta W$  between the upper and lower arms. Moreover, the reference of the 0 component is used to regulate the internal energy *W* of the converter (see [52,53] for more details on the MMC control using a bilinear *dq*0 model). These reference calculations were carried out by the energy balance control and the energy control, respectively. The reference for the *q* component was chosen as zero to diminish power losses, in this case, without loss of generality [53].

To enable grid-supporting capabilities to a GFL converter, a slight modification to the block diagram shown in Figure 3 was necessary. Based on measurements of the grid's frequency  $\omega_g$  and voltage  $v_{abc}$  and on the active and reactive powers exchanged with the grid, a contribution for supporting the frequency  $\Delta P$  or for supporting the voltage  $\Delta Q$ were obtained in the grid-supporting block and were sent as a correction to the active and reactive power references  $P^*$  and  $Q^*$ . Based on this grid-supporting control strategy, it is possible to implement future ancillary services such as virtual inertia and fast frequency response [58].



Figure 3. Block diagram of an MMC-based GFL converter embedded with supporting capabilities.

When the MMC operates according to the GFL strategy (in grid-feeding or gridsupporting modes), the control system needs a measurement of the grid phase angle  $\theta_g$ to establish the dq frame for the AC dynamics and the dq0 frame for the inner converter dynamics. Moreover, the control system also needs a measure of the amplitude of the voltage. This is provided by the PLL block shown in Figures 2 and 3. In strong grids, the PLL exhibits a satisfactory performance as the conditions of the grid change slowly. However, because of the ongoing transformation of the power system, grids are getting weaker and with lower levels of inertia that could turn the PLL into sources of instability [15], limiting the integration of GFL converters in the power system to assure the stability and the reliability of the power supply [19]. As an alternative to increase the penetration of converter interfaced generation, the GFM described in the next subsection was proposed.

#### 2.4. Grid Forming (GFM)

Although a precise definition of GFM converters is still not a consensus among the scientific community, a GFM converter can be considered as a converter that can operate and support the operation of a power grid under normal and emergency conditions without relying on services provided by synchronous generators [20]. A GFM can control directly its voltage amplitude, frequency, and phase without relying on the strength of the grid. It can provide services such as (but not limited to) black start capability, virtual inertia, primary frequency support, and islanded operation capability [59].

A block diagram of the control system of a MMC controlled following a GFM strategy is shown in Figure 4. In the GFM concept, the active and reactive power flows are controlled indirectly through the regulation of the AC voltage amplitude and the converter phase angle. When using this strategy, the AC current controller receives its references from an upper-level controller that now targets tracking the *d* and *q* components of the PCC voltage, provided by a voltage droop control. By controlling the amplitude of the PCC voltage, it is possible to regulate the reactive power exchange between the converter and the electric grid [60].



Figure 4. Block diagram of an MMC-based GFM converter.

The grid-forming strategy block is responsible for determining the phase angle  $\theta_{gf}$  of the GFM converter and its frequency. The angle  $\theta_{gf}$  is the integral over time of the frequency  $\omega_{gf}$ . So, by changing the frequency  $\omega_{gf}$ , it is possible to control the active power flow by the phase difference between the converter angle and the grid phase angle [60]. This eliminates the need for a PLL for synchronizing with the electric grid.

When using a GFM strategy, the converter frequency  $\omega_{gf}$  is no longer a frequency value associated to a real rotating mass as is the case for synchronous generators. Hence, in a 100% converter-fed power system, the meaning of frequency ceases to be related to the speed of rotating masses. Additionally, the *dq* and *dq*0 frames needed for the system are established by the converter phase angle  $\theta_{gf}$  in a GFM strategy, restricting the use of the PLL for eventual measurements of frequency.

Depending on the application, a secondary control layer can be embedded in the GFM controller to regulate voltage and frequency to their reference values. Secondary control has the task of providing references for the inner control loops of the primary control. They improve the parallel operation of the generation units and remove steady-state errors from lower control layers. In this way, the frequency and voltage references are calculated by the secondary control, such that, the power sharing and the load supply are accurate [61]. A secondary control loop can be implemented via linear techniques, such that the frequency deviation is eliminated. Therefore, a proportional plus integral term ( $\alpha_{\omega}$ ) is introduced to improve the convergence speed of frequency and to eliminate the steady state error. The control can be developed as follows:

$$\alpha_{\omega} = K_p^{\omega}(\omega^* - \omega_g) + K_i^{\omega} \int (\omega^* - \omega_g) dt$$
(1)

where  $\omega_g$  is the grid frequency, and  $\omega^*$  is the reference of frequency.  $K_p^{\omega}$  and  $K_i^{\omega}$  are the proportional and integral gains of a PI controller and can be allocated according to grid requirements. Therefore, frequency ROCOF and frequency deviation are limited [62].

To assure the elimination of deviations in a steady state and improving the rate of change of frequency (ROCOF), the secondary control term introduced in (1) is inserted in the frequency droop equation given by

$$\omega_r = \omega^* - m_p (P_m - P^*) + \alpha_\omega \tag{2}$$

where  $\omega_r$  is the frequency reference from the droop scheme,  $m_p$  is the droop coefficient,  $P_m$  is the measured active power, and  $P^*$  is the active power reference.

The same consideration of the secondary control can be adopted for voltage regulation, where an integral term given by  $\alpha_v$  can be inserted in the voltage droop. Therefore, the voltage droop control considering  $\alpha_v$  from the secondary control layer can be written as:

$$V_r = V^* - m_q (Q_m - Q^*) + \alpha_v$$
(3)

where  $V_r$  is the voltage reference resulted from droop scheme, and  $V^*$  is the rated voltage reference.  $Q_m$  is the measured reactive power,  $Q^*$  is the reactive power reference, and  $\alpha_v$  is given by:

$$\alpha_v = K_p^v(V^* - V_r) + K_i^v \int (V^* - V_r) dt$$

$$\tag{4}$$

The gains  $K_p^v$  and  $K_i^v$  are calculated according to the desired convergence speed of the secondary control layer [63].

In the following subsections, the two most prominent GFM schemes are discussed.

#### 2.4.1. Virtual Synchronous Machine

The virtual synchronous machine (VSM) provides transient power sharing and primary control frequency support independently, only based on local measurements. A VSM can also be implemented with no need of PLL, being used just for sensing the grid frequency or during initial machine starting. Moreover, the swing equation of the VSM allows the converter to interact with the grid frequency, influencing its behavior. As a result, VSMs are conceptually simple thanks to intuitive interpretation as synchronous machines' responses [22,64,65]. The VSM is implemented to provide voltage reference output where the power flow is related to inertia emulation and the angle from the swing equation, while voltage amplitude and reactive power control are made separately by the modulation index in the converter. The VSM scheme implemented in a point-to-point MMC-HVDC link is introduced in Figure 5.



Figure 5. VSM general control scheme applied to an MMC-HVDC link.

As detailed in [22,23], the VSM is implemented by using the swing equation of a synchronous machine in the VSC control structure as

$$\dot{\tilde{\omega}} = \frac{1}{M_v} [P_{ref} - P_{ele} - D_v \tilde{\omega}]$$
(5)

where  $\tilde{\omega} = \omega_{vsm} - \omega_g$  is the frequency deviation,  $\omega_{vsm}$  is the VSM's frequency,  $M_v$  is the virtual inertia coefficient, and  $D_v$  is the damping factor.  $P_{ref}$  is the active power droop reference, and  $P_{ele}$  is the measured power into the AC grid.

The virtual inertia coefficient is defined in [60]:

$$M_v = \frac{J\omega_o^2}{2S_{nom}} \tag{6}$$

where  $S_{nom}$  is the nominal apparent power of the VSC converter,  $\omega_o$  is the nominal grid frequency value, and *J* is the emulated moment of inertia.

#### 2.4.2. Droop-Based Topology

The droop control is a well-known approach used for power-sharing without necessary communication among the distributed generation units, which contributes for an easy

application. The designed control loop is composed by P - f and Q - V droops considering an electrical grid with inductive impedance (X >> R) and a large amount of inertia, which is the case for conventional power systems with high voltage transmission lines. In the classical case, Equations (2) and (3) model the droop relation. The droop control strategy has only steady-state properties, with no dynamical contribution to frequency or voltage regulation. That can result in a slow transient response with improper active power sharing. In addition, droop control is not able to bring the system back to the original equilibrium point of the system without secondary controllers [66,67]. The droop control scheme implemented in a point-to-point MMC-HVDC link is introduced in Figure 6.



Figure 6. Droop general control scheme applied to an MMC-HVDC link.

Another approach to provide a dynamical contribution to droop control is to insert a delay in the active power response that emulates the inertial behavior of a synchronous machine [68]. However, the virtual inertial contribution in droop-controlled converters is rather small [6]. In droop control applications, the measured output power is filtered to avoid noise and high-frequency components from power converter switching. Usually, a low-pass filter with a suitable time constant is applied, so the filter will induce a slower behavior in active and reactive power that can be compared with the inertial behavior of a synchronous machine [69,70]. A standard low-pass filter for active power can be described in the frequency domain:

$$P_{out}^{*}(s) = \frac{1}{1+sT_d} P_{ele}(s)$$
<sup>(7)</sup>

where  $P_{out}^*$  is the filtered output active power measured in the system, and  $T_d$  is the filter time constant.

According to [65], applying the filter dynamics (7) in the frequency droop Equation (2), we may result in the following expression that presents a virtual inertial component:

$$\frac{T_d}{R_d}\frac{d\omega}{dt} = P_m - P_{out}^* - \frac{1}{R_d}(\omega - \omega^*)$$
(8)

where  $R_d$  is the droop coefficient that is typically chosen in the range from 2 to 8%, according to the requirement of the power grid. The resulting equation of the droop control has an equivalent structure to the inertial response (5), which results in an LPF with analogous function of a virtual inertia approach. It is necessary to correctly tune the parameters of the droop regulator to obtain a small signal behavior of a synchronous machine [65].

#### 3. Modeling of a Multi-Machine, Multi-Converter Power System

Synchronous machines were the foundation of the power system for over 100 years. Because of this fact, their modeling, control, and stability were fairly well investigated. The However, the energy transition at hand is changing the foundation of the power system. Growing levels of converter-interfaced generation, such as HVDC systems and renewables, are connected each year to power systems around the globe. Therefore, the modeling, control, and stability of the power converters are now one of the main focuses of attention for researchers (see [2,3,34–37,78–80]). Specifically, stability analysis using energy functions were provided in [34,37] for single-machine, single-converter systems and in [2,3] for multi-machine, multi-HVDC systems.

In this section, we discuss the model of a multi-machine, multi-HVDC system suitable for performing stability analysis based on energy functions.

#### 3.1. Generators and Converters Models

In this work, the classical machine model [60,72] was chosen for the synchronous generators as

$$\theta_{g,i} = \omega_{g,i} \tag{9}$$

$$M_{g,i}\dot{\omega}_{g,i} = P_{mec,g,i} - P_{ele,g,i} - D_{g,i}(\omega_{g,i} - \omega_{grid})$$
(10)

where  $\theta_{g,i}$  is the rotor angle,  $\omega_{g,i}$  the frequency,  $M_{g,i}$  the inertia constant,  $P_{mec,g,i}$  the mechanical power,  $P_{ele,g,i}$  the electrical power,  $D_{g,i}$  the damping coefficient of the *i*-th synchronous generator, and  $\omega_{grid}$  is the grid's frequency.

Assuming that the control of the terminal voltage and of the internal dynamics of the MMC following a GFM strategy is achieved, then the MMC-HVDC can be modeled for the grid stability study in a similar way as a synchronous machine from the grid's perspective. When the VSM GFM strategy is adopted, the converter phase angle and frequency can be modeled as

$$\theta_{v,j} = \omega_{v,j} \tag{11}$$

$$M_{v,j}\dot{\omega}_{v,j} = P_{ref,v,j} - P_{ele,v,j} - D_{v,j}(\omega_{v,j} - \omega_{grid})$$
(12)

where  $\theta_{v,j}$  is the phase angle,  $\omega_{v,j}$  the frequency,  $M_{v,j}$  the virtual inertia coefficient,  $P_{ref,v,j}$  the power reference,  $P_{ele,v,j}$  the electrical power, and  $D_{v,j}$  the damping coefficient of the *j*-th MMC controlled as a VSM-GFM converter.

The model for a GFM droop-controlled MMC-HVDC when assuming that the voltage and internal energy dynamics controls were achieved is given by:

$$\dot{\theta}_{d,k} = \omega_{d,k} \tag{13}$$

$$\frac{T_{d,k}}{R_{d,k}}\dot{\omega}_{d,k} = P_{ref,d,k} - P_{ele,d,k} - \frac{1}{R_{d,k}}(\omega_{d,k} - \omega_{grid})$$
(14)

where  $\theta_{d,k}$  is the phase angle of the droop controlled GFM converter,  $\omega_{d,k}$  the frequency,  $P_{ref,d,k}$  the converter power reference,  $P_{ele,d,k}$  the electrical power, while  $T_{d,k}$  is the power measurement first-order filter time constant,  $R_{d,k}$  is the droop coefficient of the *k*-th MMC controlled as a droop GFM converter, and  $\omega_{grid}$  is the system's frequency.

At this moment, it is possible to trace a comparison between the two GFM control strategies. The dynamic behavior of the GFM converter frequencies given by (12) and (14) presents a similar structure as the synchronous generator swing Equation (10). Therefore, both GFM strategies are able to present some virtual inertia support that is a desirable ancillary service in the future of the power system, as reported in [81]. However, the droop-controlled GFM converter virtual inertia is dependent on the relation  $T_{d,k}/R_{d,k}$  of the first-order filter that was added to the measurement of  $P_{ref,d,k} - P_{ele,d,k}$  and the droop coefficient  $R_{d,k}$  [6,65,70]. Since the time constant is usually chosen in the order of 10 ms and the droop coefficient in the range of 3 to 8%, the resulting virtual inertia of the droop controlled GFM converter is limited to the order of a tenth of a second, while traditional synchronous generators have inertia constants in the range of 1 to 10 s.

#### 3.2. Center-of-Inertia Formulation of a Multi-Machine, Multi-GFM Converter Power System

When performing stability studies using energy functions for a multi-machine, multi-GFM converter power system, it is necessary to establish a reference to compare the rotor and converter phase angles that are essential to assert system stability. A possible reference is the center of inertia (COI). The COI is a notion of the mean motion of the system [73]. This reference is defined by the COI angle  $\theta_o$ , the COI frequency  $\omega_o$ , and the total inertia of the system  $M_T$ .

The COI for a multi-machine, multi-GFM converter comprising *L* synchronous generators, *M* MMC-HVDCs controlled according to a VSM GFM strategy, and *N* MMC-HVDCs controlled according to a droop GFM strategy is defined in the following:

$$\theta_{o} = \frac{1}{M_{T}} \left[ \sum_{i=1}^{L} M_{g,i} \theta_{g,i} + \sum_{j=1}^{M} M_{v,j} \theta_{v,j} + \sum_{k=1}^{N} \frac{T_{d,k}}{R_{d,k}} \theta_{d,k} \right]$$
(15)

$$\omega_{o} = \frac{1}{M_{T}} \left[ \sum_{i=1}^{L} M_{g,i} \omega_{g,i} + \sum_{j=1}^{M} M_{v,j} \omega_{v,j} + \sum_{k=1}^{N} \frac{T_{d,k}}{R_{d,k}} \omega_{d,k} \right]$$
(16)

$$M_T = \sum_{i=1}^{L} M_{g,i} + \sum_{j=1}^{M} M_{v,j} + \sum_{k=1}^{N} \frac{T_{d,k}}{R_{d,k}}$$
(17)

where  $M_T$  is the total inertia of the system, aggregating physical and virtual inertia. The dynamic of the COI given by (18) and (19) is described as the mean motion of the system's inertia:

$$\dot{\theta}_0 = \omega_o \tag{18}$$

$$M_T \dot{\omega}_o = \left[\sum_{i=1}^{L} (P_{mec,g,i} - P_{ele,g,i}) + \sum_{j=1}^{M} (P_{ref,v,j} - P_{ele,v,j}) + \sum_{j=1}^{N} (P_{ref,d,k} - P_{ele,d,k})\right] = P_{COI}$$
(19)

where  $P_{COI}$  corresponds to the overall power imbalance on the system. By defining the relative angles to the COI:

$$\delta_{g,i} = \theta_{g,i} - \theta_o; \ \delta_{v,j} = \theta_{v,j} - \theta_o; \ \delta_{d,k} = \theta_{d,k} - \theta_o$$
(20)

and the relative frequencies as:

$$\tilde{\omega}_{g,i} = \omega_{g,i} - \omega_o; \quad \tilde{\omega}_{v,j} = \tilde{\omega}_{v,j} - \omega_o; \quad \tilde{\omega}_{d,k} = \omega_{d,k} - \omega_o \tag{21}$$

we can rewrite the synchronous generator model (9) and (10) and the GFM converter models (11) and (12) and (13) and (14) in the relative variables given by (20) and (21) considering that the grid angle and the grid frequency are given by those of the COI.

Therefore, the resulting synchronous generator model in the COI frame is given by:

$$\dot{\delta}_{g,i} = \tilde{\omega}_{g,i} \tag{22}$$

$$M_{g,i}\tilde{\omega}_{g,i} = P_{mec,g,i} - P_{ele,g,i} - D_{g,i}\tilde{\omega}_{g,i} + \frac{M_{g,i}}{M_T}P_{COI}.$$
(23)

For the HVDC converter controlled as a VSM, the resulting model is given by:

$$\dot{\delta}_{v,j} = \tilde{\omega}_{v,j} \tag{24}$$

. .

$$M_{v,j}\dot{\tilde{\omega}}_{v,j} = P_{ref,v,j} - P_{ele,v,j} - D_{v,j}\tilde{\omega}_{v,j} + \frac{M_{v,j}}{M_T}P_{COI}$$
(25)

and for the droop-controlled converter:

$$\dot{\delta}_{d,k} = \tilde{\omega}_{d,k} \tag{26}$$

$$\frac{T_{d,k}}{R_{d,k}}\dot{\omega}_{d,k} = P_{ref,d,k} - P_{ele,d,k} - \frac{1}{R_{d,k}}\tilde{\omega}_{d,k} + \frac{T_{d,k}}{R_{d,k}M_T}P_{COI}.$$
(27)

#### 4. Lyapunov-Based Stability Analysis Using Energy Functions

As seen in the previous section, a multi-machine/multi-GFM converter is modeled by a set of differential equations. A very resource-consuming method to assert the stability of a huge nonlinear dynamical systems, such as modern power systems, is to solve in time the set of differential equations that models the system and analyze their trajectories. That is a very burdening task to carry out practically. With the advancements in computing capabilities, another way to assess the stability of a dynamic system is to implement it in a software and to carry out exhaustive simulations, trying to span all possibilities and configurations for system operation, analyzing the trajectories of all variables by numerical methods, e.g., Monte-Carlo [82,83]. These methods, on the other hand, are very time consuming and might not lead to a definitive conclusion given the exponential number of combinations arising from the complexity of the systems.

A solution for determining the stability of a dynamical system without solving its differential equations or resorting to a great number of simulations is applying Lyapunov theory. There are two paths on pursuing a Lyapunov-based stability analysis: the indirect method and the direct method [74,84,85]. The indirect method consists of linearizing the nonlinear system around one operation point and obtaining its eigenvalues. If they all have negative real parts, then the system is stable in a neighborhood of the chosen operation point for linearization. In the power systems literature, this approach is referred to as "small-signal" analysis because the conclusions drawn from it are limited to small disturbances that keep the system's trajectory around that operating point chosen for linearization.

Rather than relying upon the linearization around an operating point, the direct method allows consideration of nonlinear systems. Therefore, it can be used for analyzing the stability of power systems facing large disturbances [60,86], which is known as large-signal stability. The stability analysis resides in finding a positive definite Lyapunov function W whose time derivative  $\dot{W}$  is negative definite (or negative semi-definite given some additional conditions). In the analysis of power systems and other physical systems, a Lyapunov candidate can be obtained inspired by the system's energy functions. The use of energy functions for asserting the stability of multi-machine power systems was studied for nearly a century by power engineers as its first reports date back to the 1930s in Russian and to the late 1940s in English [87]. In the next subsection, we introduce the energy functions for the multi-machine/multi-GFM converter power system.

#### 4.1. Energy Functions for Synchronous Generators and GFM Converters

For synchronous generators, a possible  $W_{g,i}(\delta_{g,i}, \tilde{\omega}_{g,i})$  function candidate widely used in the literature (see [60,71,73,74,88]) for investigating the stability of a multi-machine power system is given by:

$$W_{g,i}(\delta_{g,i},\tilde{\omega}_{g,i}) = \frac{1}{2}M_{g,i}\tilde{\omega}_{g,i}^2 - \int_{\delta_0}^{\delta} (P_{mec,g,i} - P_{ele,g,i})d\delta$$
(28)

The function (28) is called an energy function because the first term represents the kinetic energy of the rotating rotor of the electrical machine, and the second term represents the potential energy of the generator.

For GFM converters controlled according to the VSM strategy, which results in the converter emulating the behavior of a synchronous machine, it is intuitive to propose a similar energy function  $W_{v,j}$ , where:

$$W_{v,j}(\delta_{v,j},\tilde{\omega}_{v,j}) = \frac{1}{2}M_{v,j}\tilde{\omega}_{v,j}^2 - \int_{\delta_0}^{\delta} (P_{ref,v,j} - P_{ele,v,j})d\delta.$$
(29)

For the droop GFM converter, the following energy function is proposed based on the analogies between the model of the GFM converters given by (25) and (27):

$$W_{d,k}(\delta_{dc,k},\tilde{\omega}_{d,k}) = \frac{1}{2} \frac{T_{d,k}}{R_{d,k}} \tilde{\omega}_{d,k}^2 - \int_{\delta_o}^{\delta} (P_{ref,d,k} - P_{ele,d,k}) d\delta.$$
(30)

#### 4.2. Multi-Machine, Multi-GFM Converter System Stability Analysis

When analyzing a system composed of L synchronous generators, M VSM GFM converters and N droop GFM converters, the state of the system can be defined by the relative angles and frequencies as:

$$\delta = [\delta_{g,1} \dots \delta_{g,L} \delta_{v,1} \dots \delta_{v,M} \delta_{d,1} \dots \delta_{d,N}]^T$$
(31)

\_

$$\tilde{\omega} = [\tilde{\omega}_{g,1} \dots \tilde{\omega}_{g,L} \tilde{\omega}_{v,1} \dots \tilde{\omega}_{v,M} \tilde{\omega}_{d,1} \dots \tilde{\omega}_{d,N}]^T.$$
(32)

Then, the following energy function can be proposed:

$$W(\delta,\tilde{\omega}) = \sum_{i=1}^{L} W_{g,i}(\delta_{g,i},\tilde{\omega}_{g,i}) + \sum_{j=1}^{M} W_{v,j}(\delta_{v,j},\tilde{\omega}_{v,j}) + \sum_{k=1}^{N} W_{d,k}(\delta_{d,k},\tilde{\omega}_{d,k})$$
(33)

that corresponds to adding the individual energy functions for each of the system components.

The time derivative of (33) is given by:

$$\dot{W}(\delta,\tilde{\omega}) = \sum_{i=1}^{L} \dot{W}_{g,i}(\delta_{g,i},\tilde{\omega}_{g,i}) + \sum_{j=1}^{M} \dot{W}_{v,j}(\delta_{v,j},\tilde{\omega}_{v,j}) + \sum_{k=1}^{N} \dot{W}_{d,k}(\delta_{d,k},\tilde{\omega}_{d,k})$$
(34)

where:

$$\dot{W}_{g,i} = \frac{\partial W_{g,i}}{\partial \delta_{g,i}} \frac{d\delta_{g,i}}{dt} + \frac{\partial W_{g,i}}{\partial \tilde{\omega}_{g,i}} \frac{d\tilde{\omega}_{g,i}}{dt}$$
(35)

$$\dot{W}_{v,j} = \frac{\partial W_{v,j}}{\partial \delta_{v,j}} \frac{d\delta_{v,j}}{dt} + \frac{\partial W_{v,j}}{\partial \tilde{\omega}_{v,j}} \frac{d\tilde{\omega}_{v,j}}{dt}$$
(36)

$$\dot{W}_{d,k} = \frac{\partial W_{d,k}}{\partial \delta_{g,i}} \frac{d\delta_{d,k}}{dt} + \frac{\partial W_{d,k}}{\partial \tilde{\omega}_{d,k}} \frac{d\tilde{\omega}_{d,k}}{dt}.$$
(37)

The evaluation of the properties of (35)–(37) is similar. Therefore, the step-by-step procedure is shown only for the first one. By evaluating (35), we obtain:

$$\frac{\partial W_{g,i}}{\partial \delta_{g,i}} \frac{d\delta_{g,i}}{dt} = -(P_{mec,g,i} - P_{ele,g,i})\dot{\delta}_{g,i}$$
(38)

$$\frac{\partial W_{g,i}}{\partial \tilde{\omega}_{g,i}} \frac{d\tilde{\omega}_{g,i}}{dt} = M_{g,i} \tilde{\omega}_{g,i} \dot{\tilde{\omega}}_{g,i}$$
(39)

By substituting the model (22) and (23) into the previous equations, the following partial time derivatives are obtained:

$$\frac{\partial W_{g,i}}{\partial \delta_{g,i}} \frac{d\delta_{g,i}}{dt} = -(P_{mec,g,i} - P_{ele,g,i})\tilde{\omega}_{g,i}$$
(40)

$$\frac{\partial W_{g,i}}{\partial \tilde{\omega}_{g,i}} \frac{d\tilde{\omega}_{g,i}}{dt} = \tilde{\omega}_{g,i} \left( P_{mec,g,i} - P_{ele,g,i} - D_{g,i} \tilde{\omega}_{g,i} + \frac{M_{g,i}}{M_T} P_{COI} \right). \tag{41}$$

Therefore, an expression for (35) can be obtained by adding (40) and (41) as follows:

$$\dot{W}_{g,i} = -D_{g,i}\tilde{\omega}_{g,i}^2 + \frac{M_{g,i}}{M_T}\tilde{\omega}_{g,i}P_{COI}.$$
(42)

Following the same steps, the evaluated expressions for (36) and (37) are obtained:

$$\dot{W}_{v,j} = -D_{v,j}\tilde{\omega}_{v,j}^2 + \frac{M_{v,j}}{M_T}\tilde{\omega}_{v,j}P_{\text{COI}}.$$
(43)

$$\dot{W}_{d,k} = -\frac{1}{R_{d,k}}\tilde{\omega}_{d,k}^2 + \frac{T_{d,k}}{R_{d,k}M_T}\tilde{\omega}_{d,k}P_{\text{COI}}.$$
(44)

Finally,  $\hat{W}(\delta, \tilde{\omega})$  is obtained by substituting (42)–(44) in (34):

$$\dot{W}(\delta,\tilde{\omega}) = \sum_{i=1}^{L} \left( -D_{g,i}\tilde{\omega}_{g,i}^{2} + \frac{M_{g,i}}{M_{T}}\tilde{\omega}_{g,i}P_{COI} \right) + \sum_{j=1}^{M} \left( -D_{v,j}\tilde{\omega}_{v,j}^{2} + \frac{M_{v,j}}{M_{T}}\tilde{\omega}_{v,j}P_{COI} \right) + \sum_{k=1}^{N} \left( -\frac{1}{R_{d,k}}\tilde{\omega}_{d,k}^{2} + \frac{T_{d,k}}{R_{d,k}M_{T}}\tilde{\omega}_{d,k}P_{COI} \right).$$
(45)

By rearranging the terms in (45), we obtain:

$$\dot{W}(\delta,\tilde{\omega}) = -\sum_{i=1}^{L} D_{g,i} \tilde{\omega}_{g,i}^{2} - \sum_{j=1}^{M} D_{v,j} \tilde{\omega}_{v,j}^{2} - \sum_{k=1}^{N} \frac{1}{R_{d,k}} \tilde{\omega}_{d,k}^{2} + \frac{1}{M_{T}} \left( \sum_{i=1}^{L} M_{g,i} \tilde{\omega}_{g,i} + \sum_{j=1}^{M} M_{v,j} \tilde{\omega}_{v,j} + \sum_{k=1}^{N} \frac{T_{d,k}}{R_{d,k}} \tilde{\omega}_{d,k} \right) P_{COI}$$
(46)

It follows from the definition of the COI (15)–(17) that:

$$\sum_{i=1}^{L} M_{g,i} \tilde{\omega}_{g,i} + \sum_{j=1}^{M} M_{v,j} \tilde{\omega}_{v,j} + \sum_{k=1}^{N} \frac{T_{d,k}}{R_{d,k}} \tilde{\omega}_{d,k} = 0.$$
(47)

Hence:

$$\dot{W}(\delta,\tilde{\omega}) = -\sum_{i=1}^{L} D_{g,i} \tilde{\omega}_{g,i}^2 - \sum_{j=1}^{M} D_{v,j} \tilde{\omega}_{v,j}^2 - \sum_{k=1}^{N} \frac{1}{R_{d,k}} \tilde{\omega}_{d,k}^2.$$
(48)

The resulting function  $\dot{W}(\delta, \tilde{\omega})$  is negative semi-definite; thus, the considered system state variables  $(\delta, \tilde{\omega})$  are shown to be bounded. Moreover, according to LaSalle's theorem [3,84,85], the system will asymptotically converge to the largest invariant set  $\Omega$ ,  $\Omega = \{(\delta, \tilde{\omega}) : \tilde{\omega} = 0\}$ , where  $\dot{W}(\delta, \tilde{\omega})$  in (48) is equal to zero. The set  $\Omega$  describes the points where the frequency of the system converges to the COI frequency in (15)–(17). The equilibrium point  $\delta_i^*$  of the phasor angles of each generator and converter are dependent on the power reference of each entity, as depicted in the equations in (23), (25), and (27).

Then, the multi-machine, multi-GFM converters, corresponding to the connection of multiple HVDC links with GFM converters to the legacy power system, are asymptotically stable. The analysis incorporated more than one GFM technology, which might be a possibility in the future of power systems, when HVDC terminals will be equipped with GFM control strategies to cope with the challenges of the evolving electrical system.

#### 5. Simulations and Results

To illustrate the obtained theoretical results in Section 4 for the stability of a multimachine/multi-GFM converter, the test system shown in Figure 7 was implemented in Matlab/Simulink for performing frequency response studies. This system was conceived upon a modification of the IEEE 9 bus test system by exchanging one synchronous generator by an MMC-HVDC and adding an extra bus with power generation from an HVDC system. As a result, the test network had two converter-interfaced resources and two synchronous generators (SG 1 and SG 2 shown in Figure 7).

The MMC-HVDCs considered in this work were modeled as MMC converters connected to an ideal DC voltage source. The ideal DC source was used to represent the HVDC station responsible for the DC voltage control. Both MMC-HVDCs had 500 MW of installed capacity and were controlled according to a GFM strategy. As for the synchronous generators, SG 1 had a 900 MW installed capacity and an inertia constant  $M_{g,1} = 5$  s, and SG 2 had a 500 MW installed capacity with  $M_{g,2} = 4.4$  s.

Since the GFM technology is under development, original equipment manufacturers might opt for different GFM control strategies that might better suit a specific challenge or the specifications of their products. Hence, a possible future scenario is that different GFM strategies are connected into the legacy power system during the energy transition period. To reflect these possible future alternatives, the following three scenarios were considered:

- Scenario 1: Both MMC-HVDCs follow the VSM control strategy;
- Scenario 2: Both MMC-HVDCs follow the droop control strategy;
- Scenario 3: MMC-HVDC-1 follows the VSM control strategy, and MMC-HVDC-2 follows the droop control strategy.



Figure 7. 12 bus electric grid used as the test system for the simulations.

The base dispatch scenario is shown in Table 2. Both HVDC lines were dispatched to provide 300 MW each to the system, while synchronous generator SG 1 power setpoint was 350 MW, and synchronous generator SG 2 power setpoint was 250 MW. There were, initially, three 400 MW loads (Loads 1, 2, and 3) connected to the system, which means that the non-synchronous generation instantaneous penetration in this scenario was of 50%. Then, to simulate a frequency disturbance, Load 4 rated at 450 MW was suddenly connected to the system, corresponding to approximately 38% of the system's load in the pre-disturbance scenario.

Resource	Dispatched	Resource	Load
MMC-HVDC-1	300 WM	Load 1	400 MW
MMC-HVDC-2	300 MW	Load 2	400 MW
SG 1	350 MW	Load 3	400 MW
SG 2	250 MW	Load 4	450 MW

Table 2. Base scenario of the simulation studies.

In the next subsections, the frequency stability of the test network was evaluated for the three specified scenarios when Load 4 was suddenly connected to the system.

#### 5.1. First Scenario: Two VSM-GFM Converters

In the first scenario, both HVDC stations connected into the 12-bus test system were controlled as VSMs. The virtual inertia constant for each one was chosen as  $M_{v,1,2} = 5$  s to match the highest inertia constant of the system (SG 1). Hence, the system presented a total inertia given by (17) of 19.4 s. Compared to the other scenarios analyzed in this work, Scenario 1 was the one with the highest total inertia (physical plus virtual).

Figure 8 shows the frequency of each element in the test network when Load 4 was suddenly connected at t = 2 s. It can be seen that for 4 s after the disturbance, the synchronous generators and the GFM converters present damped oscillations in their frequency. Moreover, 5 s after the disturbance, the system as a whole settled at the frequency of the COI. Hence, the test system remained stable after the frequency disturbance caused by the sudden connection of a load equivalent to 38% of the system total load.



Figure 8. Frequency behavior of the elements of the test system after disturbance in Scenario 1.

Figures 9 and 10 show the behavior of the HVDC terminals controlled according to the VSM-GFM control strategy. In Figure 9, immediately after the disturbance, the frequency reduced in both HVDC terminals, while the power delivered to the grid increased as shown in Figure 10. The converters, as per the control system design, presented the behavior of synchronous machines providing virtual inertia to the system, reducing their frequency while injecting more power into the system to counteract the disturbance.



**Figure 9.** Frequency of the voltage generated at the HVDC converter terminals after the disturbance in Scenario 1.



Figure 10. Power injected by the HVDC converter terminals after the disturbance in Scenario 1.

#### 5.2. Second Scenario: Two Droop GFM Converters

In the second scenario, both HVDC stations connected to the test system followed the droop strategy. The GFM strategy parameters corresponding to the time constant of the first order filter were chosen as  $T_{d,1,2} = 0.012$  s and the droop coefficient as  $R_{d,1,2} = 3\%$ . For this scenario, the total inertia of the system given by (17) was 10.2 s. Of all three scenarios adopted, Scenario 2 presented the lowest value of total inertia (considering both physical and virtual inertia).

The frequency of each element of the system after the disturbance corresponding to the sudden connection of Load 4 is shown in Figure 11. The contribution of the virtual inertia brought by GFM converters can be seen in Figure 11, showing that all elements of the system (both synchronous and non-synchronous power sources) had a higher frequency deviation from the reference value of 60 Hz in comparison to Figure 8. Nevertheless, the system remained stable after the disturbance.



Figure 11. Frequency behavior of the elements of the test system after disturbance in Scenario 2.

The frequency and power behavior of the converters connected into the system representing the HVDC systems are shown in Figures 12 and 13, respectively. They both injected power into the grid while their frequency was reduced to counteract the disturbance.



**Figure 12.** Frequency of the voltage generated at the HVDC converter terminals after the disturbance in Scenario 2.



Figure 13. Power injected by the HVDC converter terminals after the disturbance in Scenario 2.

#### 5.3. Third Scenario: One VSM and One Droop GFM Converter

Finally, the third scenario aimed at studying the case when the GFM technologies adopted in a system are not the same. The converter station of MMC-HVDC-1 connected into the test system was controlled as a VSM, while the MMC-HVDC-2 station followed the droop strategy. In this case, the virtual inertia constant was also taken as  $M_{v,1} = 5$  s

to match the largest synchronous generator. For the second HVDC, the time constant of the first-order filter was chosen as  $T_{d,1} = 0.012$  s, and the droop coefficient  $R_{d,1} = 0.03\%$ . For this scenario, the total inertia of the system given by (17) was 14.8 s. The total inertia concerning this scenario was intermediate, between Scenarios 1 and 2.

The frequency of the test system following the sudden connection of Load 4 is shown in Figure 14. When comparing to the frequency behavior shown in Figure 11, it is possible to verify that the frequency drop of SG 2 is improved when a larger virtual inertia is present. That is confirmed when comparing to Figure 8 where the frequency of SG 2 presented the smallest frequency nadir considering the disturbance of the sudden connection of Load 4. Additionally, in the mixed GFM technology scenario, the test system remained stable after the disturbance.



Figure 14. Frequency behavior of the elements of the test system after disturbance in Scenario 3.

Figures 15 and 16 show the frequency and the power of both GFM converters, respectively, during the studied disturbance. The frequency of both converters presented a very similar behavior. However, by analyzing Figure 16, the same frequency deviations generated two different power outputs, with the MMC-HVDC-1 converter injecting nearly 0.22 p.u. of extra power, while MMC-HVDC-2 only injected 0.05 p.u. of extra power during the frequency nadir caused by the disturbance.



**Figure 15.** Frequency of the voltage generated at the HVDC converter terminals after the disturbance in Scenario 3.



Figure 16. Power injected by the HVDC converter terminals after the disturbance in Scenario 3.

#### 5.4. The Role of Virtual Inertia

By comparing the results from Scenarios 1 through 3, it is possible to verify in the simulations the contribution of virtual inertia on the frequency support of legacy power grids. Scenario 1 presented fewer oscillations, and the converters were more responsive to frequency disturbances, followed by Scenario 3, where the GFM converter using the VSM strategy contributed more to the frequency disturbance then the droop GFM converter. As some power systems consider procuring physical (and virtual) inertia as ancillary services in a near future, operating a GFM converter with the VSM control strategy can be seen as an advantage.

#### 6. Conclusions

This work presented a Lyapunov-based stability analysis of a multi-machine/multi-GFM converters power system. A model for the legacy power system permeated with HVDC transmission systems controlled as GFM converters was introduced using the centerof-inertia formulation after a literature review on the most mature GFM technologies set the basis for the modeling of the converters. The stability analysis demonstrates that the frequencies of each element of the system are bound to converge to the frequency of the COI while including the aspect of the interoperability between different GFM converter technologies and synchronous machines.

To illustrate the theoretical results, a 12-bus test system was implemented in Matlab/Simulink to study the frequency response for a disturbance corresponding to a sudden load connection. To reflect the evolution being undergone in power systems, the test network included two synchronous generators and two MMC converter representing HVDC transmission system terminals where the GFM control strategies were implemented. Moreover, the HVDC transmission systems were considered to provide 50% of the power supply.

The simulations performed in this work confirmed the theoretical results obtained with Lyapunov theory showing that the frequency of each element of the system converges for the frequency of the COI. There were three scenarios analyzed: (i) when only VSM-GFM converters are adopted; (ii) when only droop-controlled GFM converters are adopted; and (iii) when mixed GFM technologies are used. From the analysis of these three scenarios, it was possible to verify that, independently of the chosen technology mix, the frequencies converge to the CoI one. The simulation results also show the importance of virtual inertia for the frequency support of power systems. In the considered scenarios, the ones with the highest total inertial levels (physical + virtual) presented the better frequency nadir and fewer oscillations. This was observed because of the higher initial power injection from the VSM-GFM converters from the virtual inertia response.

**Author Contributions:** Conceptualization, L.F.N.L. and G.D.; methodology, L.F.N.L. and G.D.; formal analysis, L.F.N.L., A.I. and G.D.; model development L.F.N.L., based on discussions with G.D., R.M.M., and M.B.C.S.; simulations, L.F.N.L. and F.P.; writing—original draft preparation, L.F.N.L. and F.P.; writing—review and editing, L.F.N.L., F.P., A.I., G.D., R.M.M. and M.B.C.S. All authors have read and agreed to the published version of the manuscript.

**Funding:** This research was funded by the National Council for Scientific and Technological Development (CNPq) grant number 201241/2020-0.

**Conflicts of Interest:** The authors declare no conflict of interest. The funders had no role in the design of the study; in the collection, analyses, or interpretation of data; in the writing of the manuscript; or in the decision to publish the results.

#### Abbreviations

MMC	Modular multilevel converter
HVDC	High-voltage direct-current
DC	Direct durrent
AC	Alternating current
VSM	Virtual synchronous machine
PCC	Point of common coupling
GFM	Grid forming
GFL	Grid following
PLL	Phase-locked loop
VSC	Voltage source converter
LPF	Low-pass filter
PI	Proportional-integral controller
ROCOF	Rate of change of frequency
PWM	Pulse width modulation
SG	Synchronous generator

#### References

- Liang, J.; Jing, T.; Gomis-Bellmunt, O.; Ekanayake, J.; Jenkins, N. Operation and control of multiterminal HVDC transmission for offshore wind farms. *IEEE Trans. Power Deliv.* 2011, 26, 2596–2604. [CrossRef]
- Gonzalez-Torres, J.C.; Damm, G.; Costan, V.; Benchaib, A.; Lamnabhi-Lagarrigue, F. Transient stability of power systems with embedded VSC-HVDC links: Stability margins analysis and control. *IET Gener. Transm. Distrib.* 2020, 14, 3377–3388. [CrossRef]
- Gonzalez-Torres, J.C.; Damm, G.; Costan, V.; Benchaib, A.; Lamnabhi-Lagarrigue, F. A novel distributed supplementary control of Multi-Terminal VSC-HVDC grids for rotor angle stability enhancement of AC/DC systems. *IEEE Trans. Power Syst.* 2020, 36, 623–634. [CrossRef]
- 4. Milano, F.; Dörfler, F.; Hug, G.; Hill, D.J.; Verbič, G. Foundations and challenges of low-inertia systems. In Proceedings of the 2018 Power Systems Computation Conference (PSCC), Dublin, Ireland, 11–15 June 2018; IEEE: Piscataway, NJ, USA, 2018; pp. 1–25.
- 5. Meegahapola, L.; Sguarezi, A.; Bryant, J.S.; Gu, M.; Conde, D.E.R.; Cunha, R. Power system stability with power-electronic converter interfaced renewable power generation: Present issues and future trends. *Energies* 2020, *13*, 3441. [CrossRef]
- 6. Lasseter, R.H.; Chen, Z.; Pattabiraman, D. Grid-forming inverters: A critical asset for the power grid. *IEEE J. Emerg. Sel. Top. Power Electron.* **2019**, *8*, 925–935. [CrossRef]
- 9 August 2019 Power Outage Report; Technical Report; OFGEM: UK, 2019. Available online: https://www.ofgem.gov.uk/sites/ default/files/docs/9\_august\_2019\_power\_outage\_report.pdf (accessed on 15 October 2021).
- Oni, O.E.; Davidson, I.E.; Mbangula, K.N. A review of LCC-HVDC and VSC-HVDC technologies and applications. In Proceedings of the 2016 IEEE 16th International Conference on Environment and Electrical Engineering (EEEIC), Florence, Italy, 7–10 June 2016; IEEE: Piscataway, NJ, USA, 2016; pp. 1–7.
- 9. Jovcic, D. High Voltage Direct Current Transmission: Converters, Systems and DC Grids; John Wiley & Sons: Hoboken, NJ, USA, 2019.
- Lesnicar, A.; Marquardt, R. An innovative modular multilevel converter topology suitable for a wide power range. In Proceedings of the 2003 IEEE Bologna Power Tech Conference Proceedings, Bologna, Italy, 23–26 June 2003; IEEE: Piscataway, NJ, USA, 2003; Volume 3, p. 6.
- Marquardt, R. Modular Multilevel Converter: An universal concept for HVDC-Networks and extended DC-Bus-applications. In Proceedings of the 2010 International Power Electronics Conference-ECCE ASIA, Sapporo, Japan, 21–24 June 2010; IEEE: Piscataway, NJ, USA, 2010; pp. 502–507.
- 12. Glinka, M.; Marquardt, R. A new AC/AC multilevel converter family. IEEE Trans. Ind. Electron. 2005, 52, 662–669. [CrossRef]
- 13. Peralta, J.; Saad, H.; Dennetière, S.; Mahseredjian, J.; Nguefeu, S. Detailed and averaged models for a 401-level MMC–HVDC system. *IEEE Trans. Power Deliv.* 2012, 27, 1501–1508. [CrossRef]
- 14. Van Hertem, D.; Gomis-Bellmunt, O.; Liang, J. HVDC Grids: For Offshore and Supergrid of the Future; John Wiley & Sons: Hoboken, NJ, USA, 2016.
- 15. Hatziargyriou, N.; Milanovic, J.; Rahmann, C.; Ajjarapu, V.; Canizares, C.; Erlich, I.; Hill, D.; Hiskens, I.; Kamwa, I.; Pal, B.; et al. Definition and classification of power system stability revisited & extended. *IEEE Trans. Power Syst.* **2020**, *36*, 3271–3281.
- 16. Teodorescu, R.; Liserre, M.; Rodriguez, P. *Grid Converters for Photovoltaic and Wind Power Systems*; John Wiley & Sons: Hoboken, NJ, USA, 2011; Volume 29.

- 17. Yazdani, A.; Iravani, R. Voltage-Sourced Converters in Power Systems: Modeling, Control, and Applications; John Wiley & Sons: Hoboken, NJ, USA, 2010.
- 18. Ali, Z.; Christofides, N.; Hadjidemetriou, L.; Kyriakides, E.; Yang, Y.; Blaabjerg, F. Three-phase phase-locked loop synchronization algorithms for grid-connected renewable energy systems: A review. *Renew. Sustain. Energy Rev.* **2018**, *90*, 434–452. [CrossRef]
- 19. Collados-Rodriguez, C.; Cheah-Mane, M.; Prieto-Araujo, E.; Gomis-Bellmunt, O. Stability analysis of systems with high VSC penetration: Where Is the limit? *IEEE Trans. Power Deliv.* **2019**, *35*, 2021–2031. [CrossRef]
- Matevosyan, J.; Badrzadeh, B.; Prevost, T.; Quitmann, E.; Ramasubramanian, D.; Urdal, H.; Achilles, S.; MacDowell, J.; Huang, S.H.; Vital, V.; et al. Grid-forming inverters: Are they the key for high renewable penetration? *IEEE Power Energy Mag.* 2019, 17, 89–98. [CrossRef]
- 21. Unruh, P.; Nuschke, M.; Strauß, P.; Welck, F. Overview on grid-forming inverter control methods. *Energies* 2020, *13*, 2589. [CrossRef]
- D'Arco, S.; Suul, J.A.; Fosso, O.B. Small-signal modeling and parametric sensitivity of a virtual synchronous machine in islanded operation. *Int. J. Electr. Power Energy Syst.* 2015, 72, 3–15. [CrossRef]
- D'Arco, S.; Suul, J.A.; Fosso, O.B. A Virtual Synchronous Machine implementation for distributed control of power converters in SmartGrids. *Electr. Power Syst. Res.* 2015, 122, 180–197. [CrossRef]
- 24. Zhong, Q.C.; Weiss, G. Synchronverters: Inverters that mimic synchronous generators. *IEEE Trans. Ind. Electron.* 2010, 58, 1259–1267. [CrossRef]
- 25. Zhong, Q.C.; Ma, Z.; Ming, W.L.; Konstantopoulos, G.C. Grid-friendly wind power systems based on the synchronverter technology. *Energy Convers. Manag.* 2015, *89*, 719–726. [CrossRef]
- Aouini, R.; Marinescu, B.; Kilani, K.B.; Elleuch, M. Synchronverter-based emulation and control of HVDC transmission. *IEEE Trans. Power Syst.* 2015, *31*, 278–286. [CrossRef]
- Hart, P.J.; Lasseter, R.H.; Jahns, T.M. Coherency identification and aggregation in grid-forming droop-controlled inverter networks. *IEEE Trans. Ind. Appl.* 2019, 55, 2219–2231. [CrossRef]
- Rokrok, E.; Qoria, T.; Bruyere, A.; Francois, B.; Zhang, H.; Belhaouane, M.; Guillaud, X. Impact of grid-forming control on the internal energy of a modular multilevel converter. In Proceedings of the 2020 22nd European Conference on Power Electronics and Applications (EPE'20 ECCE Europe), Lyon, France, 7–11 September 2020; IEEE: Piscataway, NJ, USA, 2020; pp. 1–10.
- 29. Guo, J.; Chen, Y.; Liao, S.; Wu, W.; Wang, X.; Guerrero, J.M. Low-Frequency Oscillation Analysis of VSMs-Based VSC-HVDC Systems Based on the Five-Dimension Impedance Stability Criterion. *IEEE Trans. Ind. Electron.* **2021**. [CrossRef]
- 30. Rokrok, E.; Qoria, T.; Bruyere, A.; Francois, B.; Guillaud, X. Classification and dynamic assessment of droop-based grid-forming control schemes: Application in HVDC systems. *Electr. Power Syst. Res.* **2020**, *189*, 106765. [CrossRef]
- 31. Raza, M.; Peñalba, M.A.; Gomis-Bellmunt, O. Short circuit analysis of an offshore AC network having multiple grid forming VSC-HVDC links. *Int. J. Electr. Power Energy Syst.* **2018**, *102*, 364–380. [CrossRef]
- 32. Tian, S.; Campos-Gaona, D.; A Lacerda, V.; Torres-Olguin, R.E.; Anaya-Lara, O. Novel control approach for a hybrid grid-forming HVDC offshore transmission system. *Energies* **2020**, *13*, 1681. [CrossRef]
- Sánchez-Sánchez, E.; Prieto-Araujo, E.; Gomis-Bellmunt, O. The role of the internal energy in MMCs operating in grid-forming mode. *IEEE J. Emerg. Sel. Top. Power Electron.* 2019, *8*, 949–962. [CrossRef]
- Fu, X.; Sun, J.; Huang, M.; Tian, Z.; Yan, H.; Iu, H.H.C.; Hu, P.; Zha, X. Large-signal stability of grid-forming and grid-following controls in voltage source converter: A comparative study. *IEEE Trans. Power Electron.* 2020, 36, 7832–7840. [CrossRef]
- 35. Pan, D.; Wang, X.; Liu, F.; Shi, R. Transient stability of voltage-source converters with grid-forming control: A design-oriented study. *IEEE J. Emerg. Sel. Top. Power Electron.* **2019**, *8*, 1019–1033. [CrossRef]
- 36. He, X.; Geng, H. Transient Stability of Power Systems Dominated by Different Types of Grid-Forming Devices. *arXiv* 2021, arXiv:2104.10849.
- Shuai, Z.; Shen, C.; Liu, X.; Li, Z.; Shen, Z.J. Transient angle stability of virtual synchronous generators using Lyapunov's direct method. *IEEE Trans. Smart Grid* 2018, 10, 4648–4661. [CrossRef]
- 38. Ackermann, T.; Ellis, A.; Fortmann, J.; Matevosyan, J.; Muljadi, E.; Piwko, R.; Pourbeik, P.; Quitmann, E.; Sorensen, P.; Urdal, H.; et al. Code shift: Grid specifications and dynamic wind turbine models. *IEEE Power Energy Mag.* **2013**, *11*, 72–82. [CrossRef]
- 39. Vandoorn, T.; De Kooning, J.; Meersman, B.; Vandevelde, L. Review of primary control strategies for islanded microgrids with power-electronic interfaces. *Renew. Sustain. Energy Rev.* **2013**, *19*, 613–628. [CrossRef]
- Han, H.; Hou, X.; Yang, J.; Wu, J.; Su, M.; Guerrero, J.M. Review of power sharing control strategies for islanding operation of AC microgrids. *IEEE Trans. Smart Grid* 2015, 7, 200–215. [CrossRef]
- 41. Rocabert, J.; Luna, A.; Blaabjerg, F.; Rodriguez, P. Control of power converters in AC microgrids. *IEEE Trans. Power Electron.* **2012**, 27, 4734–4749. [CrossRef]
- 42. Debnath, S.; Qin, J.; Bahrani, B.; Saeedifard, M.; Barbosa, P. Operation, control, and applications of the modular multilevel converter: A review. *IEEE Trans. Power Electron.* **2014**, *30*, 37–53. [CrossRef]
- 43. Raju, M.N.; Sreedevi, J.; Mandi, R.P.; Meera, K. Modular multilevel converters technology: A comprehensive study on its topologies, modelling, control and applications. *IET Power Electron.* **2019**, *12*, 149–169. [CrossRef]
- Antonopoulos, A.; Angquist, L.; Nee, H.P. On dynamics and voltage control of the modular multilevel converter. In Proceedings of the 2009 13th European Conference on Power Electronics and Applications, Barcelona, Spain, 8–10 September 2009; IEEE: Piscataway, NJ, USA, 2009; pp. 1–10.

- 45. Saad, H.; Guillaud, X.; Mahseredjian, J.; Dennetiere, S.; Nguefeu, S. MMC capacitor voltage decoupling and balancing controls. *IEEE Trans. Power Deliv.* **2014**, *30*, 704–712. [CrossRef]
- 46. Sharifabadi, K.; Harnefors, L.; Nee, H.P.; Norrga, S.; Teodorescu, R. Design, Control, and Application of Modular Multilevel Converters for HVDC Transmission Systems; John Wiley & Sons: Hoboken, NJ, USA, 2016.
- Bergna, G.; Berne, E.; Egrot, P.; Lefranc, P.; Arzande, A.; Vannier, J.C.; Molinas, M. An energy-based controller for HVDC modular multilevel converter in decoupled double synchronous reference frame for voltage oscillation reduction. *IEEE Trans. Ind. Electron.* 2012, 60, 2360–2371. [CrossRef]
- 48. Bergna-Diaz, G.; Suul, J.A.; D'Arco, S. Energy-based state-space representation of modular multilevel converters with a constant equilibrium point in steady-state operation. *IEEE Trans. Power Electron.* **2017**, *33*, 4832–4851. [CrossRef]
- Bergna-Diaz, G.; Freytes, J.; Guillaud, X.; D'Arco, S.; Suul, J.A. Generalized voltage-based state-space modeling of modular multilevel converters with constant equilibrium in steady state. *IEEE J. Emerg. Sel. Top. Power Electron.* 2018, 6, 707–725. [CrossRef]
- Prieto-Araujo, E.; Junyent-Ferré, A.; Collados-Rodríguez, C.; Clariana-Colet, G.; Gomis-Bellmunt, O. Control design of Modular Multilevel Converters in normal and AC fault conditions for HVDC grids. *Electr. Power Syst. Res.* 2017, 152, 424–437. [CrossRef]
- Dadjo Tavakoli, S.; Prieto-Araujo, E.; Sánchez-Sánchez, E.; Gomis-Bellmunt, O. Interaction assessment and stability analysis of the MMC-based VSC-HVDC link. *Energies* 2020, 13, 2075. [CrossRef]
- 52. Vatani, M.; Hovd, M.; Saeedifard, M. Control of the modular multilevel converter based on a discrete-time bilinear model using the sum of squares decomposition method. *IEEE Trans. Power Deliv.* **2015**, *30*, 2179–2188. [CrossRef]
- de Oliveira, G.C.; Damm, G.; Monaro, R.M.; Lourenco, L.F.; Carrizosa, M.J.; Lamnabhi-Lagarrigue, F. Nonlinear control for modular multilevel converters with enhanced stability region and arbitrary closed loop dynamics. *Int. J. Electr. Power Energy Syst.* 2021, 126, 106590. [CrossRef]
- 54. Antonopoulos, A.; Ängquist, L.; Harnefors, L.; Ilves, K.; Nee, H.P. Global asymptotic stability of modular multilevel converters. *IEEE Trans. Ind. Electron.* **2013**, *61*, 603–612. [CrossRef]
- 55. Harnefors, L.; Antonopoulos, A.; Ilves, K.; Nee, H.P. Global asymptotic stability of current-controlled modular multilevel converters. *IEEE Trans. Power Electron.* 2014, *30*, 249–258. [CrossRef]
- Saad, H.; Peralta, J.; Dennetiere, S.; Mahseredjian, J.; Jatskevich, J.; Martinez, J.; Davoudi, A.; Saeedifard, M.; Sood, V.; Wang, X.; et al. Dynamic averaged and simplified models for MMC-based HVDC transmission systems. *IEEE Trans. Power Deliv.* 2013, 28, 1723–1730. [CrossRef]
- 57. Sánchez-Sánchez, E.; Prieto-Araujo, E.; Junyent-Ferré, A.; Gomis-Bellmunt, O. Analysis of MMC energy-based control structures for VSC-HVDC links. *IEEE J. Emerg. Sel. Top. Power Electron.* **2018**, *6*, 1065–1076. [CrossRef]
- 58. Miller, N.; Lew, D.; Piwko, R. Technology capabilities for fast frequency response. GE Energy Consult. Tech. Rep. 2017, 3, 10–60.
- 59. Meegahapola, L.; Mancarella, P.; Flynn, D.; Moreno, R. Power system stability in the transition to a low carbon grid: A techno-economic perspective on challenges and opportunities. *Wiley Interdiscip. Rev. Energy Environ.* **2021**, *10*, e399. [CrossRef]
- 60. Kundur, P. Power System Stability and Control. Stab. Control; McGraw-Hill: New York, NY, USA, 1994.
- 61. Rebours, Y.G.; Kirschen, D.S.; Trotignon, M.; Rossignol, S. A survey of frequency and voltage control ancillary services—Part I: Technical features. *IEEE Trans. Power Syst.* **2007**, *22*, 350–357. [CrossRef]
- 62. Lee, J.O.; Kim, Y.S. Frequency and State-of-Charge Restoration Method in a Secondary Control of an Islanded Microgrid without Communication. *Appl. Sci.* 2020, *10*, 1558. [CrossRef]
- Dohler, J.S.; de Almeida, P.M.; de Oliveira, J.G. Droop Control for Power Sharing and Voltage and Frequency Regulation in Parallel Distributed Generations on AC Microgrid. In Proceedings of the 2018 13th IEEE International Conference on Industry Applications (INDUSCON), Sao Paulo, Brazil, 12–14 November 2018; IEEE: Piscataway, NJ, USA, 2018; pp. 1–6.
- 64. Tamrakar, U.; Shrestha, D.; Maharjan, M.; Bhattarai, B.P.; Hansen, T.M.; Tonkoski, R. Virtual inertia: Current trends and future directions. *Appl. Sci.* 2017, 7, 654. [CrossRef]
- D'Arco, S.; Suul, J.A. Virtual synchronous machines—Classification of implementations and analysis of equivalence to droop controllers for microgrids. In Proceedings of the 2013 IEEE Grenoble Conference, Grenoble, France, 16–20 June 2013; IEEE: Piscataway, NJ, USA, 2013; pp. 1–7.
- 66. De Brabandere, K.; Bolsens, B.; Van den Keybus, J.; Woyte, A.; Driesen, J.; Belmans, R. A voltage and frequency droop control method for parallel inverters. *IEEE Trans. Power Electron.* **2007**, *22*, 1107–1115. [CrossRef]
- 67. Mohd, A.; Ortjohann, E.; Morton, D.; Omari, O. Review of control techniques for inverters parallel operation. *Electr. Power Syst. Res.* **2010**, *80*, 1477–1487. [CrossRef]
- 68. Arani, M.F.M.; Mohamed, Y.A.R.I.; El-Saadany, E.F. Analysis and mitigation of the impacts of asymmetrical virtual inertia. *IEEE Trans. Power Syst.* **2014**, *29*, 2862–2874. [CrossRef]
- 69. Soni, N.; Doolla, S.; Chandorkar, M.C. Improvement of transient response in microgrids using virtual inertia. *IEEE Trans. Power Deliv.* 2013, *28*, 1830–1838. [CrossRef]
- 70. D'Arco, S.; Suul, J.A. Equivalence of virtual synchronous machines and frequency-droops for converter-based microgrids. *IEEE Trans. Smart Grid* **2013**, *5*, 394–395. [CrossRef]
- 71. Pai, M. Power System Stability: Analysis by the Direct Method of Lyapunov; Elsevier: Amsterdam, The Netherlands, 1981.
- 72. Pai, M.; Sauer, P.W. Stability analysis of power systems by Lyapunov's direct method. *IEEE Control Syst. Mag.* **1989**, *9*, 23–27. [CrossRef]

- 73. Pai, M. Energy Function Analysis for Power System Stability; Springer Science & Business Media: Berlin, Germany, 2012.
- 74. Machowski, J.; Lubosny, Z.; Bialek, J.W.; Bumby, J.R. *Power System Dynamics: Stability and Control*; John Wiley & Sons: Hoboken, NJ, USA, 2020.
- 75. Willems, J.L.; Willems, J.C. The application of Lyapunov methods to the computation of transient stability regions for multimachine power systems. *IEEE Trans. Power Appar. Syst.* **1970**, *5*, 795–801. [CrossRef]
- 76. Willems, J. Direct method for transient stability studies in power system analysis. *IEEE Trans. Autom. Control* **1971**, *16*, 332–341. [CrossRef]
- 77. Bergen, A.R.; Hill, D.J. A structure preserving model for power system stability analysis. *IEEE Trans. Power Appar. Syst.* **1981**, *1*, 25–35. [CrossRef]
- Ramasubramanian, D.; Vittal, V.; Undrill, J.M. Transient stability analysis of an all converter interfaced generation WECC system. In Proceedings of the 2016 Power Systems Computation Conference (PSCC), Genoa, Italy, 20–24 June 2016; IEEE: Piscataway, NJ, USA, 2016; pp. 1–7.
- Ramasubramanian, D.; Yu, Z.; Ayyanar, R.; Vittal, V.; Undrill, J. Converter model for representing converter interfaced generation in large scale grid simulations. *IEEE Trans. Power Syst.* 2016, 32, 765–773. [CrossRef]
- 80. Rokrok, E.; Qoria, T.; Bruyere, A.; Francois, B.; Guillaud, X. Transient Stability Assessment and Enhancement of Grid-Forming Converters Embedding Current Reference Saturation as Current Limiting Strategy. *IEEE Trans. Power Syst.* 2021. [CrossRef]
- 81. *Review of Relevance of Current Techniques to Advanced Frequency Control;* Technical Report; RESERVE: 2017. Available online: http://www.re-serve.eu/files/reserve/Content/Deliverables/D2.2.pdf (accessed on 15 October 2021).
- 82. Li, W. Reliability Assessment of Electric Power Systems Using Monte Carlo Methods; Springer Science & Business Media: Berlin, Germany, 2013.
- 83. Liu, W.; Guo, D.; Xu, Y.; Cheng, R.; Wang, Z.; Li, Y. Reliability assessment of power systems with photovoltaic power stations based on intelligent state space reduction and pseudo-sequential Monte Carlo simulation. *Energies* **2018**, *11*, 1431. [CrossRef]
- 84. Khalil, H.K. Nonlinear Systems, 3rd ed.; Patience Hall: Englewood Cliffs, NJ, USA, 2002.
- 85. Slotine, J.J.E.; Li, W. Applied Nonlinear Control; Prentice Hall: Englewood Cliffs, NJ, USA, 1991; Volume 199.
- Kundur, P.; Paserba, J.; Ajjarapu, V.; Andersson, G.; Bose, A.; Canizares, C.; Hatziargyriou, N.; Hill, D.; Stankovic, A.; Taylor, C.; et al. Definition and classification of power system stability IEEE/CIGRE joint task force on stability terms and definitions. *IEEE Trans. Power Syst.* 2004, 19, 1387–1401.
- 87. Fouad, A. Transient stability criteria. In Proceedings of the 1975 IEEE Conference on Decision and Control including the 14th Symposium on Adaptive Processes, Houston, TX, USA, 10–12 December 1975; IEEE: Piscataway, NJ, USA, 1975; pp. 309–314.
- 88. Fouad, A.A.; Vittal, V. *Power System Transient Stability Analysis Using the Transient Energy Function Method*; Pearson Education: London, UK, 1991.